#### **ORIGINAL RESEARCH**



# European Banking Union and bank risk disclosure: the effects of the Single Supervisory Mechanism

Yener Altunbaş<sup>1</sup> · Salvatore Polizzi<sup>2</sup> · Enzo Scannella<sup>2</sup> · John Thornton<sup>3</sup>

Accepted: 26 June 2021 © The Author(s) 2021

## Abstract

This paper provides evidence on the impact of European Banking Union (BU) and the associated Single Supervisory Mechanism (SSM) on the risk disclosure practices of European banks. The onset of BU and the associated rules are considered as an exogenous shock that provides the setting for a natural experiment to analyze the effects of the new supervisory arrangements on bank risk disclosure practices. A Difference-in-Differences approach is adopted, building evidence from the disclosure practices of systemically important banks supervised by the European Central Bank (ECB) and other banks supervised by national regulators over the period 2012-2017. The main findings are that bank risk disclosure increased overall following BU but there was a weakening of disclosure by SSM-supervised banks relative to banks supervised by national authorities. We also find that the overall positive effect of the BU on bank disclosure is stronger for less profitable banks and in the most troubled economies of the Eurozone (GIPSI countries), while the negative effect on centrally supervised banks is stronger if bank CEOs act also as chairmen (CEO duality). We interpret these findings in light of the fact that the new institutional arrangements for bank supervision under which the ECB relies on local supervisors to collect the information necessary to act gives rise to inefficiencies with respect to the speed and completeness of the information flow between SSM supervised banks and the ECB, which are reflected in bank disclosure practices.

**Keywords** Risk disclosure · Banks · Banking union · Single supervisory mechanism · Principal-agent problem

Salvatore Polizzi salvatore.polizzi@unipa.it

Yener Altunbaş y.altunbas@bangor.ac.uk

Enzo Scannella enzo.scannella@unipa.it

John Thornton John.Thornton@uae.ac.uk

- <sup>1</sup> Business School, Bangor University, Bangor, UK
- <sup>2</sup> Department of Economics, Business and Statistics, University of Palermo, Palermo, Italy
- <sup>3</sup> US Department of the Treasury, Business School, University of East Anglia, Norwich, UK

JEL Classification G21 · G28 · G38

# 1 Introduction

This paper aims to investigate the effects that the change in banking supervision and the associated rules of the Banking Union (BU) in Europe has had on the disclosure practices of banks. Banks are notoriously opaque because of their risk-taking and maturity transformation role, which makes them difficult to assess without considerable information on their financial position and risk-taking practices (Morgan 2002; Flannery et al. 2013). If risk disclosure is adequate, it can serve as an outside mechanism for monitoring the behavior of senior management (Eng and Mak 2003) and facilitate access to external finance at a reasonable cost of capital (Botosan 1997; Botosan and Plumlee 2002; Easley and O'Hara 2004; Cheng et al. 2006; Zhang and Ding 2006; Eaton et al. 2007; Kothari et al. 2009). However, voluntary risk disclosures are not always optimal for several reasons. First, disclosures are costly and banks weigh the costs against the potential benefit of lower funding costs when deciding on the quality, level and frequency of disclosures. Second, banks have considerable discretion in deciding on the information to be disclosed, which also means that voluntary disclosures may not necessarily generate optimal outcomes. Landsman (2006) notes that banks may be able to affect disclosure content through the timing of earnings announcements and balance sheet adjustments. Gunther and Moore (2003) show that US banks in poorer financial condition are more likely to understate financial losses. Finally, even if risk disclosure is optimal, the regulatory architecture may mitigate against regulators taking the appropriate action. For example, the literature shows that supervisors have diverging interests when the discharge of supervisory duties affects economies asymmetrically. This may be the case, for example, when the affiliates of a cross-border banking group are of diverging importance for their respective national financial markets (e.g., Holthausen and Rønde 2004; Bolton and Oehmke 2016). In the context of European BU, the potentially diverging interests of supervisors could have important implications as they could result in suboptimal information sharing and cooperation between the European Central Bank and the national bank regulators (see, for example, Carletti, et al. 2015, 2021).

Supervision and bank disclosure are closely related. The existing literature provides evidence that bank supervision has a strong impact on bank disclosure (Barakat and Hussainey 2013; Mester 2017; Costello et al. 2018). Bank supervisory authorities are essential to maintain the integrity and transparency of the whole banking sector. Furthermore, bank disclosure is extremely beneficial for supervisors. Nier and Baumann (2006) show that more transparency decreases equity return volatility, and consequently improve supervisors' view of the risk and relative performance of the bank. Higher levels of transparency can also supplement conventional supervisory tools, as the increasing complexity of large financial institutions makes them difficult to control using traditional monitoring techniques (Flannery 2001).

Bank opacity and inadequate risk disclosure have been cited as an important factor contributing to the financial crisis (Avgouleas 2009; Gorton 2009; Sowerbutts et al. 2013) in that banks did not report enough information about the risks that they were exposed to. The lack of disclosure by banks magnified uncertainty about the underlying value of assets and off-balance sheet exposures to structured credit products, which led to considerable reluctance by counterparties to trade and further fueled the market turmoil. Gorton and Ordoñez (2014) show how opacity can prevent market participants from distinguishing high-risk from low-risk institutions, with market participants supplying funds to both healthy and unsound banks on similar terms and conditions. These information problems can increase financial fragility.

In response to the financial crisis, financial regulators in major jurisdictions have taken several steps aimed at increasing bank risk disclosures to reduce the likelihood and severity of future crises and promote financial stability. In the European Union (EU), the risk disclosure practices of European banks had been largely harmonized by the requirement that banks (and other firms) report according to International Financial Reporting Standards beginning in 2005. Following the crisis, three further reforms were introduced. First, in 2014 Basel III Pillar 3 disclosures requirements were introduced into EU law through the Capital Requirements Directive IV (CRD IV), which introduced standardized regulatory reporting by specifying the information firms must report to supervisors in areas such as own funds, large exposures and financial information.<sup>1</sup> Second, the European BU project began to be implemented from late 2014 after the fragmented system of national supervision and bank resolution was viewed as having been characterized by regulators pursuing domestic objectives that exacerbated systemic risk (Draghi 2018). A key pillar of BU is the two-tier single supervisory mechanism (SSM) for banks. The SSM regulation comprises an institutional architecture that sees the European Central Bank (ECB) as the supranational hub at the center of the supervisory network that also assigns important supervisory functions to the national regulator spokes (National Competent Authorities, NCAs). At the hub, banks deemed "significant" are supervised directly by the ECB, though on the basis of information reported by banks to their national regulator and subsequently passed on to the ECB.<sup>2</sup> The supervision of significant banks includes the ECB conducting stress tests on banks and subjecting them to capital or risk limits and changes in management if problems are found.<sup>3</sup>In the spokes, smaller banks continue to be directly monitored and supervised by their national authorities. The effective functioning of the SSM therefore hinges critically on making the interfaces within the SSM work smoothly and, in particular, on extensive information sharing and cooperation between the ECB and NCAs. The third reform, was the introduction of the Markets in Financial Instruments Directive II in 2018, which imposed more reporting requirements on all EU banks.

The BU and the new disclosure rules were expected to facilitate closer monitoring of banks by regulators and other stakeholders thereby subjecting bank managers to greater discipline and making the financial system more stable. A central supervisory authority would concern itself with stability and welfare at systemic-wide level, above any specific local interests, whereas local supervisors would have specific interests pertaining to the geographic area under their supervision that might be detrimental for the systemic-wide financial stability (Draghi 2018). However, the reliance of the ECB on national regulators

<sup>&</sup>lt;sup>1</sup> The reform has also embraced subsequent revisions to the disclosure requirements aimed at addressing perceived shortcomings in Pillar 3. In this regard, it is important to notice that 2014 Basel III Pillar 3 disclosures requirements have had an important impact on the disclosure provided by European banks, but exclusively in the Pillar 3 disclosure report, which is provided separately from the annual financial reports (Scannella and Polizzi 2021).

 $<sup>^2</sup>$  See Tröger (2014) for a detailed description of the division of labor between the ECB and the NCAs in the SSM.

<sup>&</sup>lt;sup>3</sup> A bank was deemed significant if: assets exceeding  $\notin$ 30 billion; assets exceeded both  $\notin$ 5 billion and 20% of the GDP of the member state in which it is located; it was among the three most significant banks of the country in which it is located; it had large cross-border activities; and it bank received, or had applied for, assistance from euro-zone bailout funds.

to pass-on information reported by significant banks has raised doubts about the legitimacy of the ECB's supervisory role. As Carletti et al. (2021) note, an institutional design in which a centralized agency has full power over all decisions regarding banks but relies on local supervisors to collect the information necessary to act entails a principal-agent problem between the central and local supervisors if their objective functions differ. As a result of different objective functions, information collection may be inferior to that under either fully independent local supervisors or under centralized information collection and may increase risk-taking by regulated banks. Thus, whether a "tougher" central supervisor—the ECB—would result in more optimal bank disclosure is an open question.

In this paper, we investigate the impact of BU (and the SSM in particular) and the associated Pillar 3 disclosure rules on the risk disclosure activities of EU banks. We treat the onset of BU as an exogenous shock that provides the setting for a natural experiment to analyze the effects of the new supervisory arrangements on bank risk disclosure practices. To this end, we use content analysis techniques and construct an expert-validated dictionary to analyze bank annual reports and exploit the introduction of BU and the new rules. We then employ Difference-in-Differences (DiD) estimation where banks subject to the SSM act as the treatment group and banks that remain subject to supervision by the national authority act as the control group. We test two hypotheses in our analysis. The first hypothesis is that the onset of BU resulted in an overall increase in risk disclosure by European banks ('significant' or otherwise). The second hypothesis is that the additional requirements of the SSM decreased risk disclosure by SSM supervised banks (treated entities) relative to that by other banks. To preview the main findings, we find support for the first hypothesis in that risk disclosure by all banks appears to have increased in a number of key categories following the onset of BU and the new disclosure rules. Our results also support the second hypothesis as risk disclosure by SSM supervised banks appears to have worsened in comparison to nationally supervised banks, despite the increase in disclosure overall. We attribute this second finding to inefficiencies associated with the two-tier supervisory mechanism under BU, which are reflected in bank risk disclosure. In order to solve these problems, we suggest that the BU be reformed such that SSM supervised banks report directly to the ECB rather than the present arrangement whereby the ECB relies on national regulators for information on these banks. In an additional analysis, we also find that the overall positive effect of the BU on bank disclosure is stronger for less profitable banks and in the most troubled economies of the Eurozone (GIPSI countries), while the negative effect on SSM supervised banks is stronger in banks where the CEO is also the chairman (CEO duality).

Our study is motivated by the importance of understanding the effects of the new banking supervision structure in Europe on bank disclosure practices. Although the extant literature has shown that banking supervision and disclosure are tightly related, to the best of our knowledge, this is the first study to examine the effects on risk disclosure of the European BU and the associated SSM.

The main contribution of the paper is to extend the limited empirical literature on the effects of BU on bank behavior. Studies of this type to date have focused on the impact on equity prices, credit contagion, market credibility, and public finance implications. We believe that this is the first empirical study on the impact of BU on bank risk disclosure practices. Second, we contribute to the discussion on the effects of bank supervision on bank disclosure practices. More generally, our study extends the limited European banking literature employing content analysis of annual reports.

The remainder of the paper is organized as follows. Section 2 describes the theoretical framework and develops the research hypotheses. Section 3 reviews the related literature on

risk disclosure. Section 4 discusses the methodology, identification strategy, and data. Section 5 reports the empirical results and Sect. 6 concludes.

## 2 Theoretical framework and research hypotheses

The literature offers various theoretical frameworks to explain the effect of banking supervision on bank disclosure practices, and their analysis is important to understand the impact of the BU on bank risk disclosure. In our research setting, we draw upon: (i) the organization society theories (Burgstahler and Dichev 1997; Cho et al. 2015; Cohen et al. 2017) that shed light into the effects of banking supervision on bank disclosure; and (ii) the theories that analyze the effectiveness of the banking supervision in a multi-supervisor setting (Agarwal et al. 2014; Carletti et al. 2021), which provide an explanation on whether or not the supervisory function performed by the SSM is more effective than that of the NCAs.

The organization-society theories (which include stakeholder theory, resource dependence theory and legitimacy theory) (Burgstahler and Dichev 1997; Cho et al. 2015; Cohen et al. 2017) represent a fundamental theoretical framework to explain the role that banking supervisors play in shaping bank disclosure practices (Barakat and Hussainey 2013). These theories posit that banking supervisory authorities that are fully informed and able to promptly and precisely take corrective actions, are effective and influential in motivating bank managers to provide comprehensive disclosure. There are two main conditions essential for supervisors to carry out their supervisory activities effectively: (i) the capability of taking corrective actions to promote changes, and; (ii) the speed and precision of taking these decisions. While both local and central supervisors should be capable of taking corrective action, when it comes to the speed and precision of deciding how to behave, there is a substantial difference between them. Given that the current design of the supervisory mechanism requires NCAs to act as information collector on behalf of the SSM, the central supervisor may not be able to promptly take supervisory corrective actions. This indirect and hence slower flow of information may also result in an information loss for the ultimate supervisor. In addition, given that NCAs have been the sole supervisors until 2014, they may be more precise in taking the right corrective actions, as they can rely also on a more complete set of soft information about the entities they supervise. For these reasons, the central supervisor would be in no position to fulfil its role effectively. On closer inspection, the legitimacy theory (O'Donovan 2002; Cho and Patten 2007; Cho et al. 2015) posits that firms in general and bank in particular may comply with disclosure requirements, in order to confirm and show their full adherence to the institutional values of the society (i.e. to gain institutional legitimacy). The resource dependence theory (Boyd 1990; Cohen et al. 2017) postulates that organization survival and growth depends on some important resources that are available in the external environment. Thus, firms compete with each other in order to control these resources. In Europe, the supervisory authorities can grant and withdraw banking licenses.<sup>4</sup> The license for the banking activity is a crucial resource for financial institutions and banks can use disclosure as a means to convince the supervisor that the bank continues to merit a license. The stakeholder theory (Bowen et al. 1995; Burgstahler and Dichev 1997) posits that stakeholders have various expectations and influence on the firm. Thus, banks may want to provide an effective disclosure to interact and

<sup>&</sup>lt;sup>4</sup> https://www.bankingsupervision.europa.eu/banking/tasks/authorisation/html/index.en.html.

communicate more effectively with the most influential stakeholders, of which the supervisor is amongst the most important. However, if the central supervisor is not capable of quickly and precisely taking corrective actions, its monitoring function may not be sufficiently effective. A slow, imprecise and not fully informed supranational supervisor may not be able to achieve a high level of institutional legitimacy, show that it controls valuable resources for financial institutions, and that it is an influential stakeholder. In our research setting, given that the SSM still has to rely on the information provided by the NCAs, it may be less informed, slower and less precise, resulting in an inefficient supervisory function. This inefficiency will be reflected also in bank disclosure, resulting in a less comprehensive disclosure for SSM supervised banks in comparison to the financial institutions monitored by national supervisors, after the establishment of the BU.

Given that the current supervisory system in Europe requires the cooperation of national and supranational supervisors, the theories that study the effectiveness of banking supervision in a multi-supervisor setting represent an important point of reference for our research setting (Agarwal et al. 2014; Carletti et al. 2021). Agarwal et al. (2014) analyze the supervisory decisions of US bank regulators, exploiting a legally determined rotation policy, which assigns state or federal supervisors to the same financial institution at predetermined time periods. The authors show that different supervisors implement the same rules inconsistently, due to differences in their utility functions. More specifically, local supervisors carry out their activities in a softer way during stressed economic conditions, because tough supervision might increase the likelihood of bank failures and lead to a loss of local lending activity and banking employment. In contrast, federal supervisors are more concerned about systemic stability at supranational level, rather than about the geographical distribution of banking credit supply and employment. According to this 'local interest hypothesis', a central supervisor should perform better than local supervisors, as they do not have specific interests in favour of large banks at local level. In contrast, local supervisors have specific interests on the geographic area under their supervision. Furthermore, local supervisors compete with each other. For instance, they may want to attract banks from closer geographical areas or avoid that their local banks move elsewhere. In order to do so, they could perform a less demanding supervisory function. This circumstance may give financial institutions the chance to exploit a regulatory arbitrage from different jurisdictions. The findings of Agarwal et al. (2014) are crucial to understand the trade-offs related to the distribution of supervisory responsibility and powers across different authorities. The authors themselves argue that their findings should be taken into consideration for the debate concerning the redesign of European bank supervisory system. Although NCAs might have an informational advantage, their utility functions are crucial in determining the efficacy of the supervisory mechanism. In particular, NCAs may want to be softer with distressed banks, if they are too big to fail for their national economy. However, from a theoretical perspective, there are various reasons why the current European supervisory system may not work fully effectively. NCAs still have room to influence the supervisory process carried out by the SSM, given that the latter still relies on the information provided by the former in order to perform its supervisory function. Consequently, these diverging interests between national and central supervisors may result in a lack of effectiveness of the supervisory mechanism. The conflicting utility functions and objectives of the local supervisors may still result in an insufficient effectiveness of the SSM, as long as NCAs can strongly interfere in the supervisory process. Hence, the SSM may be in no position to fulfil its supervisory function effectively. In light of the postulates of the aforementioned organization society theories, this lack of effectiveness may be reflected in the disclosure provided by the banks under SSM supervision.

Carletti et al. (2021) study the behaviour of supervisors in the so called 'hub-andspokes' regime: where a central supervisor has juridical power over all decisions concerning financial institutions, but it relies upon local supervisors in order to collect the information it needs to take the necessary corrective actions. This institutional structure entails a double principal agent problem, between the bank and its supervisor, and between the local and the central supervisor. The latter is a serious issue, because the two supervisors have different utility functions and final objectives. In this situation, the scale of information collection of local supervisors will not only be inferior to that of a central supervisor, which directly collects all information, but also to that of independent local supervisors, which remains inferior to the central supervisor model. The reason behind this suboptimal level of information collection lies on the fact that local supervisors prefer to remain ignorant, rather than learning information that could lead the central supervisory authority to actions that are against their interests. This, in turn, leads to an inefficient supervisory system and to a less sound banking sector. The authors themselves state that their theoretical framework is inspired by the current supervisory design in Europe. This model suggests that unless the spokes (NCAs) and the hub (the ECB) act jointly with the same objectives, the effectiveness of the whole supervisory system will be compromised, suggesting that the current supervisory structure in Europe is far from being able to solve the problems of the fragmentation of bank monitoring activity. In conclusion, in our research setting, this model would predict a worsening in the supervisory function, and, consequently, a less comprehensive bank disclosure for SSM supervised financial institutions after the establishment of the BU. In addition, the European Directive 2013/36 and the Regulation (EU) N. 575/2013 prescribe several national options and discretions that can be applied on the basis of certain national circumstances.<sup>5</sup>

However, there should not be any fear of regulatory and supervisory inconsistency for the financial institutions that are still supervised by the NCAs. These financial institutions are not affected by the aforementioned 'hub-and-spokes' regime, as they are supervised by national authorities. Furthermore, after the establishment of the BU, an important message has been sent to the entire banking system: banking supervision and risk disclosure have become a priority at national and international level. Hence, our expectation is that there may be an overall improvement in the supervisory system across Europe, after the establishment of the BU. In addition, after the BU in 2014, the workload of the local supervisory authorities has decreased, as NCAs are no longer responsible for the supervision of the largest financial institutions. Hence, they have more time and resources to monitor the other financial institutions. Based on these considerations and theories, our first research hypothesis is the following:

Hypothesis 1 Overall, the banking system has improved its risk disclosure after the BU.

In contrast, according to the aforementioned theories, the establishment of the BU may result in less comprehensive disclosure specifically for SSM supervised banks, in comparison to NCA monitored financial institutions, because of the aforementioned problems of the 'hub-and-spokes' regime. Furthermore, given that the European Directive 2013/36 and the Regulation (EU) N. 575/2013 prescribe several national options and discretions, there might be a fear of regulatory and supervisory inconsistencies for the financial institutions supervised by the SSM (Carboni et al. 2017). This would support the idea that flexible,

<sup>&</sup>lt;sup>5</sup> See Regulation (EU) N. 575/2013, Article 400(2)(c) for further information.

well informed and quick local supervisors are better than a slow, less informed and potentially inconsistent central supervisor. Hence, we develop our second research hypothesis as follows:

**Hypothesis 2** The banks supervised by the SSM have worsened their risk disclosure in comparison to NCA monitored financial institutions after the establishment of the BU.

Both of our research hypotheses are related to the theoretical perspectives proposed by the agency theory (Jensen and Meckling 1976), which represents a suitable framework to analyze the relationships between national and supranational banking authorities within the BU (Nielsen and Smeets 2018; Carletti et al. 2021), and bank disclosure behaviour (Barakat and Hussainey 2013; Al-Hadi et al. 2016). For example, Nielsen and Smeets (2018) analyze the role of EU and national institutions in guiding the reform process of the European Monetary Union. These authors rely on the principal-agent theory to explain how these authorities have cooperated to set up the BU. In contrast, with reference to bank disclosure, Al-Hadi et al. (2016) analyze the impact of corporate governance mechanisms on bank disclosures under the lenses of the agency theory. In particular, the agency theory posits that increased managerial monitoring is positively associated with risk disclosure and bank transparency.

## 3 Evidence on the determinants of risk disclosure

Failures in bank-risk disclosure emerged as a troublesome feature of the financial crisis. Sowerbutts et al. (2013, p. 327) state that "In plain terms, banks did not report enough information about the assets they were holding or the risks that they were exposed to." Publishing better information could reduce the probability of future financial crises by making supervision more effective and sudden changes in sentiment of outsider stakeholders less likely. The relevant empirical literature has several strands and much of it relies on content analysis of annual reports and financial statements and relates to non-financial institutions.<sup>6</sup> One set of studies examines how firm-specific characteristics impact risk-disclosure. For example, Abraham and Cox (2007) find that risk reporting in the annual reports of US corporates is positively related to governance (larger board size and a higher proportion of independent directors) and negatively related to the proportion of institutional share ownership. Linsley and Shrives (2006) report that in a sample of 79 UK company annual reports larger firms with greater environmental risk have higher risk disclosures. Finally, Helbok and Wagner (2006) find that disclosure of operational risk is higher in banks with lower capital-asset and profitability ratios in a sample of 142 banks from North America, Asia, and Europe.

A second set of studies focuses on firms' incentives to disclose risk (Nissim 2003). For example, Ball et al. (2003) examine the interaction between accounting standards and the incentives of managers and auditors in East Asian countries whose standards derive from common law sources that are widely viewed as being of a higher quality than code law standards. They show that financial reporting quality is not higher than under code law and attribute differences to the low quality of report preparers. Ball and Shivakumar (2005) find that market demand for information results in UK private company financial

<sup>&</sup>lt;sup>6</sup> See, for example, Weber (1990) and Krippendorff (2004) for discussions of content analysis.

reporting being of a lower quality than that for public companies, even though both types face broadly equivalent regulations. Miihkinen (2012) finds that firm size, profitability, and foreign listing status were important determinants of disclosure in a sample of listed Finnish firms. Finally, Goncharenko et al. (2018) present a model of the effect of information disclosure on banks' portfolio risk and show that information disclosure lowers the expected risk-adjusted profits for banks, especially for systemically important institutions.

Studies specifically on the impact of regulatory changes on firms' risk disclosure practices are quite rare. Woods et al. (2008) find that the increasing adoption of international accounting standards and global convergence of accounting regulations has done little to reduce the diversity of market risk disclosure practices at international and European level. Barakat and Hussainey (2013) report that powerful and independent bank supervisors mitigate the incentives for entrenched bank executives to withhold voluntary disclosure. Barakat and Hussainey's (2013) work belongs to a developed strand of literature that analyzes the relationship between bank supervision and bank disclosure practices and level of transparency (Mester 2017; Costello et al. 2018). This stream of literature provides evidence that bank supervisory authorities are essential to maintain the integrity and transparency of the whole banking sector. Higher levels of transparency can also supplement conventional supervisory tools, as the increasing complexity of large financial institutions makes them difficult to control using traditional monitoring techniques (Flannery 2001). Burks et al. (2018) use the relaxation of interstate branching restrictions in the US to examine how increases in competition affected voluntary disclosure through press releases. They report that more competition is associated with an increase in press releases that become more negative in tone as entry barriers decrease, and that disclosures by public banks and by banks issuing equity become incrementally positive in tone when entry barriers decrease. They interpret this as the increase in disclosure being aimed at deterring entry via negative information, which is mitigated by an incentive to communicate positive information to investors. Finally, Kleymenova and Zhang (2019) investigate how the Dodd-Frank Act (DFA) affected voluntary disclosures of large bank holding companies relative to other banks and unregulated firms in the financial sector. Using a difference-in-differences research design, they find that following the introduction of the DFA, large banks become less likely to issue earnings forecasts containing bad news, reduce the frequency of issuing earnings forecasts but increase the frequency of providing forecasts for dividends and return on assets; in addition, large banks provided less information than other banks on regulated activities and instead focused on financial performance and market innovation.

Empirical literature specifically on the effects of Banking Union is especially scarce. To date, there have been studies of the impact of BU on bank credit risk (Avignone et al. 2020), on the equity prices of banks to be subjected to the SSM mechanism (Carboni et al. 2017; Sahin and de Haan 2016), on bank and sovereign credit contagion (Sáiz et al. 2019), and on the market credibility (Pancotto, 2019), systemic implications (Hüser et al. 2017), and public finance aspects (Benczur et al. 2017) of the BU's bank resolution regime. As far as we are aware, there are no empirical studies of the impact of BU and the SSM on the risk disclosure practices of European banks.

## 4 Methodology and data

#### 4.1 Dependent variable

There is substantial evidence that content analysis techniques are trustworthy tools to extract valuable information from financial reports (Li 2010; Brown and Tucker 2011; Bushman et al. 2016; Andrés et al., 2021). To investigate bank disclosure practices, we create a tailored dictionary designed to analyze bank reports and financial statements, as applying standardized dictionaries outside the context for which they were created might invalidate the analysis (Loughran and McDonald 2011; Beattie 2014; Kearney and Liu 2014). We create the dictionary by selecting the most relevant words to test our research hypotheses by drawing on a selection of specialized banking and finance dictionaries<sup>7</sup> (Fitch 2018; Rutherford 2013; Shim and Constas 2016; Law and Smullen 2018). The dictionary has been subsequently validated by a panel of experts from the ECB and the academia that were asked to suggest additional words, eliminate those that were not considered relevant and provide suggestions on the categorization.<sup>8</sup> The procedure resulted in a dictionary of 125 words, aggregated into four different categories, which are shown in Table 1. The category "risk management disclosure" consists of terms that financial institutions are supposed to use to describe the risk management, monitoring, and measurement procedures and functions they adopt for the wide range of risks they are exposed to. The category "risk exposure disclosure" comprises the words that provide information related to the vulnerability of the bank to these risks.<sup>9</sup> The category "references to the regulatory framework" is a list of terms that identify the most important regulatory and supervisory authorities that influence European banks' activities at the international level. The category "reassuring disclosure" consists of terms that financial institutions may want to use to reassure stakeholders about the bank's financial position, performance and risk exposure.

For each category of words, we created a disclosure index (DI) computed as the standardized mean of the relative occurrences of each word belonging to the category,<sup>10</sup> divided by the total number of words of the entire document, as suggested by previous disclosure studies (Tetlock et al. 2008; Bushman et al. 2016). Formally:

$$DI_{Catk} = \frac{mow_{Catk} - \mu_{Cat k}}{\sigma_{Catk}}$$
(1)

in which:  $mow_{Cat k}$  is the mean of the relative occurrences of the words belonging to the generic category k, and  $\mu$  and  $\sigma$  are respectively the mean and the standard deviation.<sup>11</sup>

<sup>&</sup>lt;sup>7</sup> Although other disclosure indexes based on qualitative content analysis methodologies have been proposed in the literature (Scannella and Polizzi 2018, 2020), they are not suited to the analysis of large samples. In addition, more sophisticated co-occurrence analyses (Illia et al., 2014) are not suited to our research questions. Hence, we rely on a quantitative content analysis methodologies based on disclosure dictionaries.

<sup>&</sup>lt;sup>8</sup> We gratefully thank Giuseppe Avignone, Alessio Reghezza and Laura Santucci for the validation of the dictionary.

<sup>&</sup>lt;sup>9</sup> The first two categories represent different aspects of the same risk disclosure concepts.

<sup>&</sup>lt;sup>10</sup> The count of the occurrences includes stemmed words. Hence, we count the number of times these stemmed words appear in the annual report.

<sup>&</sup>lt;sup>11</sup> The standardization may be necessary if our disclosure index is non-stationary, which might happen in case there are regime changes in the word distribution (Tetlock et al. 2008). Because of this standardization, the mean and the standard deviation of the disclosure indexes are zero and one, respectively.

Risk management	Risk exposure	Regulatory framework	Reassurance
Advanced measurement approach	Ambiguity	Bank for International Set-	Bail out
Asset quality review	Bank run	tiements Descil Committee	Boom Continuous Frantino
Back test	Bankrupt Daula viala	Basel Committee	Contingency Funding
Credit rating	Bank risk		and Recovery Plan
tions	Commodity risk	European Banking Authority European Central Bank	Economic growth
Evaluation	Compliance risk	European Stability Mecha-	Institutional Protection
Expected loss	Contagion	nism	Scheme
Exposure at default	Counterparty risk	European Systemic Risk	Lender of Last Resort
Internal capital adequacy assess-	Country risk	Board	Reputation
ment	Credit risk	International Accounting	Rescue
Process	Crisis	Standard	Safe
Internal assessment	Currency risk	International Financial	Sound
Internal control	Danger	Reporting	Stability
Internal model approach	Default	Standard	Too big to fail
Internal rating based	Default risk	Basel pillar	Trust
Incremental risk charge	Emergency	Prudential regulation	
Liquidity coverage ratio	Emergency risk	Regulation	
Loss given default	Enterprise risk	Single Resolution Mechanism	
Measurement	Failure	Single Supervisory Mecha-	
Net stable funding ratio	Foreign exchange risk	nism	
Probability of default	Fraud	Supervision	
Quantitative impact study	Idiosyncratic risk	•	
Rating	Illiquid		
Reverse stress test	Incremental risk charge		
Standardized model	Insolvency risk		
Stressed value at risk	Instability		
Test	Interest rate risk		
Unexpected loss	Liquidity risk		
Valuation risk	Market risk		
Value at risk	Operational risk		
Conditional value at risk	Other risk		
Exposure at default	Panic		
Enterprise risk management	Peril		
Evaluation risk	Political risk		
Model risk	Regulatory risk		
Probability of default	Reputational risk		
Risk appetite framework	Residual risk		
Risk avoidance	Risk concentration		
Risk coverage	Risk exposure		
Risk culture	Risk factor		
Risk management	Settlement risk		
Risk measurement	Sovereign risk		
Risk mitigation	Strategic risk		
Risk monitoring	Stress		
Risk provisioning			
Risk tolerance			
Risk transfer			
Stress test			

 Table 1 Disclosure categories for content analysis of annual reports

This table is an expert-validated tailored dictionary designed to analyze bank annual reports drawing from a selection of specialized banking and finance dictionaries (Fitch 2018; Rutherford 2013; Shim and Constas 2016; Law and Smullen 2018)

These disclosure indices represent the dependent variables of our econometric models. More specifically,  $mow_{Catk}$  is defined as:

$$mow_{Catk} = \sum_{i=1}^{n} \frac{(number of occurrences word i)/n}{Total number of words of the annual report}$$
(2)

in which n is the number of words belonging to the category k.

## 4.2 Econometric methodology

To study the effects of BU and the SSM on bank risk disclosure, we employ a differencein-differences (DiD) methodology.<sup>12</sup> Formally, for each disclosure index, we estimate the following model:

$$DI_{Cat k_{ijt}} = \beta_0 + \beta_1 B U_t + \beta_2 \operatorname{Treat}_{ij} + \beta_3 (BU^* \operatorname{Treat}_{ijt}) + \gamma Con + \delta CFE + \varepsilon_{ijt}$$
(3)

in which:  $DI_{Catkijt}$  is the disclosure index of the category k for bank i in country j at time t; BU is the enactment date of Banking Union. It is equal to one after 2014 and zero otherwise. *Treat* equals zero for the control group (i.e., non-systemically important banks) and one for the treatment group (i.e., significant banks).  $\beta_0$  is the expected level of disclosure when all dependent variables are equal to zero.  $\beta_1$  captures any change in disclosure in both groups following the onset of BU.  $\beta_2$  captures disclosure differences between the treatment and control groups before BU.  $\beta_3$  represents the main coefficient of interest because it captures the effect of BU on the treatment group. More specifically, it measures the difference in disclosure from the pre-treatment to the post-treatment period between SSM supervised banks relative to banks supervised by the national regulator. A positive (negative) statistically significant sign for  $\beta_3$  would imply *ceteris paribus* that disclosure increases (reduces) more for SSM-supervised banks than for banks supervised by the national regulator.

Of the other components of the model,  $\gamma Con$  is a vector of bank- and country-specific controls widely used in the banking literature (Richardson and Welker 2001; Linsley et al. 2006; Chen and Vashishtha 2017) to detect cross-bank heterogeneity that could affect bank disclosure independently of the establishment of the BU, and  $\delta CFD$  represents country fixed effects to control for country heterogeneity. The bank-specific controls are: the ratio of equity to total assets (ETA) as measure of bank capital; the return on assets (ROA) as a measure of bank profitability; the ratio of loans to assets (LOANS) to capture credit structure; the ratio of off-balance sheet assets to total assets (OBS) to capture off-balance sheet exposure; the ratio of customer deposits to total liabilities (CUST) to capture funding structure; and the dummy variable (BIG4D) equal to unity for banks whose financial statement is audited by one of the big four audit firms and zero otherwise. The country level controls are: the World Bank's business disclosure index (WBBDI) to control for the potential effects of different regulatory frameworks for financial disclosure at country level; and a dummy variable (NSAD) equal to unity for the countries that have more than one national supervisor and zero otherwise to capture potential diverging interests between different supervisors at country level.

<sup>&</sup>lt;sup>12</sup> The DiD approach has been widely adopted in banking and accounting studies (e.g., Barth and Israeli 2013; Berger et al. 2014; Fiordelisi et al. 2017).

#### 4.3 Data

The word category variables were obtained by counting the number of occurrences of the words of our tailored dictionary in bank annual reports. The annual reports were collected manually from each bank's official website with the websites identified using the Orbis Bank Focus (Bureau van Dijk) database. The World Bank's Business Extent of Disclosure Indexes were collected from the World Bank "Doing Business" database.

The bank sample consists of 75 SSM-supervised significant banks (the treatment group) and 150 other large but less important European financial institutions supervised by national regulators (the control group). More specifically, given that size is the most important parameter taken into account by the ECB to decide upon the systemically importance of the banks of the sample, we selected the largest nationally supervised entities that provide an audited and English version of their annual report.<sup>13</sup> Balance sheet and performance data covering the period 2011-2017 are from the Orbis Bank Focus database with the balance sheet variables winsorized at the 1% and 99% level to avoid the influence of outliers. For both groups of banks, we excluded those banks that did not provide documentation for all years, and that did not provide audited and English versions of their consolidated annual financial reports. The treatment and control groups both comprise large banks that are important at the national level and all banks in the control group are of a size that they are potentially subject to the supervision of the SSM if they were to grow in size either organically or by merging with other financial institutions. The geographical distribution of the banks in the sample is shown in Table 2. The final sample period is 2012–2017, which includes 3 years before the introduction of the BU and 3 years after its onset. The period is intentionally short as the change in the treatment group should be concentrated around the onset of the treatment (Bertrand et al. 2004). As we are specifically interested in assessing the impact of the BU, the choice of this time period enhances the validity of our analysis. Descriptive statistics for all of the variables are provided in Table 3. We tested the control variables for multicollinearity problems through the Variance Inflation Factor (VIF), which provided an average VIF of 1.14 suggesting that our control variables are not highly correlated. The correlation matrix is provided in Table 4. The regressions were estimated with bank-level clusters, allowing for correlation in the error term (Petersen 2009) and robust standard errors to control for dependence and heteroscedasticity (White 1980).

# 5 Empirical results

## 5.1 Baseline results

Table 5 shows the results of the empirical analysis from estimating Eq. (3) for the four bank disclosure categories. For each category of disclosure, the coefficient of the BU onset

<sup>&</sup>lt;sup>13</sup> For additional information on the significance criteria adopted by the ECB see: https://www.bankingsup ervision.europa.eu/banking/list/criteria/html/index.en.html.

The SIFI and the less significant institutions of our sample kept their significant / less significant status throughout the sample period. Although, potentially, banks could change their significant / less significant status, the number of significant supervised entities has been very stable throughout the sample period, because the ECB is committed "to ensure that high supervisory standards are applied consistently." (https://www.bankingsupervision.europa.eu/banking/list/criteria/html/index.en.html.). For these reasons, it is not possible to control for bank fixed effect in our empirical analysis, because there would be a multi-collinearity problem with the *TREAT* dummy that identifies SSM supervised banks.

Country	Number of banks	Percent of total
Austria	10	4.44
Belgium	6	2.67
Bulgaria	5	2.22
Croatia	4	1.78
Cyprus	3	1.33
Czech Republic	5	2.22
Denmark	6	2.67
Estonia	4	1.78
Finland	4	1.78
France	18	8.00
Germany	22	9.78
Greece	5	2.22
Hungary	6	2.67
Ireland	8	3.56
Italy	23	10.22
Latvia	3	1.33
Lithuania	3	1.33
Luxembourg	7	3.11
Malta	6	2.67
Netherlands	16	7.11
Poland	8	3.56
Portugal	4	1.78
Romania	3	1.33
Slovakia	4	1.78
Slovenia	3	1.33
Spain	9	4.00
Sweden	5	2.22
United Kingdom	25	11.11
Total	225	100.00

Orbis Bank Focus (Bureau van Dijk) database

dummy is statistically significant and positive indicating that, in general, BU was associated with an increase in bank disclosure, consistent with our first research hypothesis. However, in three of the four disclosure categories (columns 1, 2 and 4) the coefficient on BU\*Treat is statistically significant and negative, which suggests a reduction in disclosures related to risk management, risk exposure, and reassurances by SSM-supervised banks *relative* to these disclosures by nationally supervised banks following the onset of BU.<sup>14</sup> This finding supports our second research hypothesis. In contrast, the coefficient on BU\*Treatis positive and statistically significant regarding disclosures that reference the regulatory framework, indicating that SSM supervised banks increased their disclosures in this regard

 Table 2
 Number of banks by

country of origin

<sup>&</sup>lt;sup>14</sup> This finding should be interpreted in light of the fact that the parameter associated with the Treat dummy is positive. This result indicates that while SSM supervised banks were generally characterized by higher levels of disclosure, their disclosure practices worsened after the establishment of the BU compared to nationally supervised banks."

#### Table 3 Descriptive statistics

	Observations	Mean	Standard deviation	Minimum	Maximum
Dependent variables (full sample)					
Risk management	1575	0.000	1.000	-1.910	9.314
Risk exposure	1575	0.000	1.000	-1.994	9.072
Regulatory framework	1575	0.000	1.000	-1.725	8.133
Reassurance	1575	0.000	1.000	-1.570	8.934
Dependent variables (treated banks)					
Risk management	525	0.148	0.913	-1.681	7.813
Risk exposure	525	0.207	0.990	-1.634	7.575
Regulatory framework	525	0.173	0.884	-1.582	6.179
Reassurance	525	0.086	0.894	-1.570	6.935
Dependent variables (control group)					
Risk management	1050	-0.074	1.032	-1.910	9.314
Risk exposure	1050	-0.103	0.989	-1.994	9.072
Regulatory framework	1050	-0.086	1.042	-1.725	8.133
Reassurance	1050	-0.043	1.046	-1.337	8.934
Independent variables (full sample)					
Bank equity to total assets (ETA)	1209	0.095	0.070	0.008	0.935
Bank profitability (ROA)	1209	0.015	0.373	-2.205	2.923
Bank loans to total assets (LOANS)	1209	0.117	0.136	0.003	0.697
Off-balance sheet assets to total assets (OBS)	1209	0.184	0.328	0.000	2.520
Customer deposits to total liabilities (CUST)	1209	0.528	0.246	0.000	0.967
Big four auditor dummy (BIG4D)	1209	0.946	0.227	0.000	1.000
Business disclosure index (WBBDI)	1209	6.330	2.206	1.000	10.000
More than one national regulator dummy (NSAD)	1209	0.499	0.531	0.000	1.000
Corporate governance variables (robustness test)	)				
CEO's age (CEOage)	1209	56.66	8.757	33.00	82.00
CEO's gender (CEOgen)	1209	0.046	0.211	0.000	1.000
CEO's duality (CEOdual)	1209	0.082	0.275	0.000	1.000
Two-tier system (2TS)	1209	0.546	0.498	0.000	1.000
Senior executives (SeniorExec)	1209	2.843	0.900	1.609	7.740

The dependent variables were obtained by counting the number of occurrences of the words of our tailored dictionary in bank annual report. The bank financial statement data are from the Orbis Bank Focus (Bureau van Dijk) database. The Business disclosure index is from the World Bank "Doing Business" database. The corporate governance variables are from Bordex, Moody's, Orbis Bank Focus (Bureau van Dijk) and Thomson Reuter Eikon

relative to nationally supervised banks. What might explain the reduced disclosure by SSM supervised banks in three of the four disclosure categories? The finding is consistent with the principal-agent problem between the ECB and the national supervisors as suggested by Carletti et al. (2021). This principal-agent problem appears to have resulted in inefficiencies with respect to the speed and completeness of the information flow between SSM supervised banks and the ECB. In this context, the information collection role played by NCAs in the current supervisory system in Europe results in a double principle-agent problem that causes inefficiencies in banking supervision, which are reflected in the disclosure

Table 4 Correlation	n matrix												
Variables	(1)	(2)	(3)	(4)	(5)	(9)	(7)	(8)	(6)	(10)	(11)	(12)	(13)
(1) ETA	1.000												
(2) ROA	0.075	1.000											
(3) LOANS	0.008	-0.008	1.000										
(4) OBS	0.070	0.001	0.045	1.000									
(5) CUST	-0.032	-0.001	-0.209	-0.010	1.000								
(6) BIG4D	-0.036	-0.009	-0.044	0.006	-0.132	1.000							
(7) WBBDI	0.028	-0.102	-0.060	-0.112	0.018	-0.063	1.000						
(8) NSAD	0.004	0.007	0.076	0.022	-0.024	-0.011	0.136	1.000					
(9) CEOage	0.017	-0.051	-0.031	0.057	-0.086	0.007	0.014	-0.146	1.000				
(10) CEOgen	-0.045	0.036	-0.067	0.012	-0.060	-0.037	-0.099	0.017	-0.149	1.000			
(11) CEOdual	-0.071	-0.028	0.076	-0.084	0.063	-0.009	0.117	0.135	-0.031	0.020	1.000		
(12) 2TS	-0.036	0.083	0.031	0.109	0.035	0.041	-0.486	0.050	-0.258	0.092	-0.004	1.000	
(13) SeniorExec	-0.118	-0.030	0.016	0.101	-0.175	0.033	-0.104	-0.199	0.117	0.039	-0.011	0.102	1.000

Y. Altunbaş et al.

Dependent variable:	(1)	(2)	(3)	(4)
-	Risk management	Risk exposure	Regulatory framework	Reassurance
Banking Union (BU)	0.2617***	0.2397***	0.1885***	0.1764***
	(0.0453)	(0.0526)	(0.0704)	(0.0415)
SSM supervised bank (Treat)	0.2069***	0.2979**	0.0702	0.1940**
	(0.0286)	(0.1454)	(0.1204)	(0.0881)
BU*Treat	-0.1733***	-0.1543***	0.1316**	-0.1811***
	(0.0388)	(0.0448)	(0.0632)	(0.0317)
ETA	-0.7145***	-0.7777**	0.1576	-0.0107
	(0.2363)	(0.3466)	(0.1575)	(0.3768)
ROA	-0.1348**	-0.1219***	-0.0324	-0.1213***
	(0.0602)	(0.0334)	(0.0537)	(0.0195)
Loans	0.2534**	0.1178	0.4009***	0.5973**
	(0.1109)	(0.1599)	(0.1303)	(0.2602)
OBS	0.0248	0.0109	$-0.1988^{***}$	-0.1276
	(0.0415)	(0.0527)	(0.0456)	(0.0967)
CUST	-0.3699**	-0.3056***	-0.0788	0.1419
	(0.1539)	(0.0765)	(0.2174)	(0.1104)
BIG4D	0.1461	0.2976**	0.4204**	0.0624
	(0.1154)	(0.1181)	(0.1997)	(0.1906)
WBBD	0.1897***	0.1083*	0.1287***	0.0855*
	(0.0735)	(0.0616)	(0.0286)	(0.0507)
NSAD	-0.6204	-0.0620	-0.7346***	-0.0934
	(0.4213)	(0.1732)	(0.2666)	(0.2532)
Observations	1,209	1,209	1,209	1,209
R <sup>2</sup>	0.302	0.365	0.373	0.287
Number of id	225	225	225	225
Clusters	Yes	Yes	Yes	Yes
Country fixed effects	Yes	Yes	Yes	Yes

Table 5 Baseline difference-in-differences estimates: dependent variable-disclosure category

The dependent variable is a bank risk disclosure index for different categories of bank risk. Robust standard errors in parenthesis. \*\*\*, \*\* and \* indicate statistical significance at the 1, 5 and 10% levels respectively

practices of SSM supervised banks. The result also supports the view expressed by Gros and Schoenmaker (2014) that the institutional arrangements under BU are the result of a compromise of national authorities that do not want to give up their political power in favor of a centralized authority. These problems cause inefficiencies in bank supervision, which are reflected in bank disclosure practices.

Of the control variables, more disclosure is provided by banks that are less capitalized and less profitable and have a higher proportion of loans on total assets, which may be because banks are more vulnerable to market shocks and credit risk and wish to allay stakeholder fears about risk exposure. These findings are also in line with the idea that when financial institutions face difficult situations, their shareholders and investors are more willing to reduce the probability of banking crises by exercising market discipline (Nier and Baumann 2006). This market discipline role can be played only if adequate levels

Dependent variable:	(1)	(2)	(3)	(4)
	Risk management	Risk exposure	Regulatory framework	Reassurance
BU fake	0.0397	-0.0405	0.0753*	-0.0201
	(0.0536)	(0.0391)	(0.0417)	(0.0419)
Treat	0.3117*	0.3826	0.0659	0.3181**
	(0.1764)	(0.2874)	(0.2021)	(0.1461)
BU fake*Treat	-0.0320	-0.0081	-0.0056	-0.0589
	(0.0515)	(0.0727)	(0.0504)	(0.0364)
Observations	586	586	586	586
R-squared	0.346	0.409	0.358	0.287
Number of id	205	205	205	205
Clusters	Yes	Yes	Yes	Yes
Country fixed effects	Yes	Yes	Yes	Yes
Controls	Yes	Yes	Yes	Yes

 Table 6
 Placebo test: dependent variable—disclosure category

The dependent variable is a bank risk disclosure index for different categories of bank risk. Robust standard errors in parenthesis. \*\*\*, \*\* and \* indicate statistical significance at the 1, 5 and 10% levels respectively

of disclosure are provided.<sup>15</sup> More specifically, the stakeholders of banks that are more vulnerable to credit risk and market shocks demand higher levels of disclosure aiming to reduce bank risk taking incentives by means of market discipline. Disclosure is also likely to be greater if banks are audited by one of the big four audit firms, because external auditors play an important role in ensuring that shareholders and potential investors are adequately informed about the risk exposure of the banks, and consequently they induce banks to provide higher levels of disclosure (Al Lawati et al. 2021). In contrast, banks whose activities are funded by a larger share of customer deposits are associated with less disclosure, probably reflecting the more stable nature of these source of funding and the reduced need to satisfy other stakeholders' concerns about risk exposure. This finding is in line with the literature that provides evidence that the market discipline role played by depositors is not always fully effective in the banking industry (Berger and Turk-Ariss 2015), or less effective compared to that of investors and shareholders.

## 5.2 Robustness tests and additional analyses

We subject our key results to several robustness tests. First, a possible concern related to estimating Eq. (3) is that bank disclosure may have been affected by events that occurred before 2014 that are not captured in the specification. For example, banks might have changed their risk disclosure practices in response to the announcement of the BU in 2012 that would take place rather than in response to its actual onset in 2014. In order to rule out this hypothesis we carry out a placebo test, widely used in previous studies that adopted the DiD identification strategy (e.g., Hertzberg et al. 2011; Balakrishnan et al. 2014; Schepens 2016). In this test, we assume that the treatment took place in 2012 instead of 2014, and

<sup>&</sup>lt;sup>15</sup> The literature suggests that higher levels of disclosure are generally associated with a decreased likelihood of bank failure (Cordella and Yeyati 1998).

Table 7         Relative percentages of the words in common between each category and cluster		Risk manage- ment (%)	Risk (%)	Regulatory framework (%)	Reas- surance (%)
	Cluster 1	28	26	20	26
	Cluster 2	0	100	0	0
	Cluster 3	17	25	28	30
	Cluster 4	21	6	73	0

The bold values show the percentage of words in common between each cluster and the category represented by it (i.e. the maximum value by row)

#### Table 8 Cluster analysis

Dependent variable	Cluster 1 risk management	Cluster 2 risk exposure	Cluster 3 reassurance	Cluster 4 regulatory framework
BU	0.3102***	0.2339***	0.2397***	0.1864***
	(0.0608)	(0.0449)	(0.0532)	(0.0552)
Treat	0.3286***	0.3137***	0.1685***	0.1871***
	(0.0341)	(0.1197)	(0.0554)	(0.0720)
BU*Treat	-0.0934***	-0.1435***	-0.1497***	0.1095**
	(0.0355)	(0.0415)	(0.0278)	(0.0556)
Observations	1,209	1,209	1,209	1,209
R-squared	0.292	0.399	0.349	0.361
Number of id	225	225	225	225
Clusters	Yes	Yes	Yes	Yes
Country fixed effects	Yes	Yes	Yes	Yes
Controls	Yes	Yes	Yes	Yes

The dependent variable is a bank risk disclosure index for different categories of bank risk. Robust standard errors in parenthesis. \*\*\*, \*\* and \* indicate statistical significance at the 1, 5 and 10% levels respectively

estimate the effect of this fictitious BU. In order to center the time horizon of the analysis around the onset of the false treatment, we analyze the time period running from 2011 to 2014 for this robustness test. The results are reported in Table 6 and show that the fictitious BU had no statistically significant effect on our variables of interest, supporting our finding that it was the onset of BU and the actual change in banking supervision arrangements that triggered the changes in banks' disclosure practices.

Second, our results are subject to the limitation that our word categorization might not be the correct way to create our disclosure categories. To rule out this hypothesis, we use an alternative purely objective statistical approach to create our disclosure categories. Specifically, we use the k-means clustering algorithm proposed by Hartigan and Wong (1979) to aggregate the words according to the variability of their occurrences in bank annual reports.<sup>16</sup> The result of this new categorization is a four-category clustering, which shares some common characteristics with our original categorization, supporting its reliability

<sup>&</sup>lt;sup>16</sup> See Krink et al. (2007) for insights on the k-means cluster methodology in a banking context.

Dependent variable:	Risk management	Risk exposure	Regulatory framework	Reassurance
BU	0.3102***	0.2339***	0.1859***	0.2397***
	(0.0608)	(0.0449)	(0.0552)	(0.0532)
Treat	0.3286***	0.3137***	0.1868***	0.1685***
	(0.0341)	(0.1197)	(0.0720)	(0.0554)
BU*Treat	-0.0934***	-0.1435***	0.1120**	-0.1497 ***
	(0.0355)	(0.0415)	(0.0556)	(0.0278)
Observations	1,209	1,209	1,209	1,209
R-squared	0.292	0.399	0.361	0.349
Number of id	225	225	225	225
Clusters	Yes	Yes	Yes	Yes
Country Fixed Effects	Yes	Yes	Yes	Yes
Controls	Yes	Yes	Yes	Yes

 Table 9
 Principal components analysis

and accuracy. Table 7 reports the relative proportion of words in common between each k-means cluster and our original categorization. The results indicate that a high proportion of the words of each cluster belong to our risk disclosure categories (28%, 100%, 30%, and 73% for clusters 1 to 4, respectively after normalization), suggesting that our original categorization is reliable. We check whether the clusters identified by the k-means approach behave similarly to our original categories by running regressions substituting our original categories with the k-means clusters. These results are reported in Table 8 and show that for each cluster, the regressions are qualitatively unchanged from those of our baseline models. Hence, we can conclude that our original categorization is reliable and accurate as it is validated by an objective statistical methodology.

Third, our results might be affected by the choice of the weights assigned to each word of the four categories we employ. As explained, the term  $mow_{Cat 1}$  in Eq. (1) is the mean of the relative occurrences of the words belonging to the first category. It is implicit that all words have the same weight equal to unity. To test the robustness of our results, we use principal component analysis to aggregate the occurrences of the words of each category into a single variable.<sup>17</sup> For each category of our dictionary, we use the first principal component as the dependent variable of our regressions. The results reported in Table 9 are consistent with the risk disclosure regressions proposed in our baseline models.

Fourth, we deal with two potential shortcomings of the DiD methodology that might cast doubt on our baseline results. First, one of the most severe assumptions of the DiD procedure is that the control group is a valid counterfactual of the treatment. Even though our sample selection approach should guarantee that our control group (nationally supervised banks) is similar to the treatment one (SSM-supervised banks), we test the robustness of our model to this assumption by selecting a restricted control group using propensity

<sup>&</sup>lt;sup>17</sup> Principal component analysis is a statistical technique that converts a set of variables into a smaller set of linearly uncorrelated variables, named principal components, which preserve the information and structure of the original variables (Cumming and Wooff 2007). This methodology has been widely used in banking and accounting studies (Larcker et al. 2007; Ludvigson and Ng 2007; Carlson and Wheelock 2018).

Dependent variable	Risk management	Risk exposure	Regulatory framework	Reassurance
BU	0.2465***	0.2151***	0.1734**	0.1610***
	(0.0452)	(0.0510)	(0.0702)	(0.0401)
Treat	0.2233***	0.3446**	0.0428	0.2303**
	(0.0509)	(0.1630)	(0.1368)	(0.1063)
BU*Treat	-0.1704***	-0.1492***	0.1458**	-0.1752***
	(0.0401)	(0.0514)	(0.0655)	(0.0348)
Observations	1,070	1,070	1,070	1,070
R-squared	0.292	0.367	0.391	0.292
Number of id	200	200	200	200
Clusters	Yes	Yes	Yes	Yes
Country fixed effects	Yes	Yes	Yes	Yes
Controls	Yes	Yes	Yes	Yes

 Table 10
 Propensity score matching

 Table 11
 Comparison of bank specific characteristics for treated and untreated banks after propensity score matching

Variables	Treated		Untreated		Difference
	Mean	Standard deviation	Mean	Standard deviation	means
ETA	0.0897	0.1138	0.0905	0.0495	-0.0007*
ROA	0.0179	0.2767	0.0169	0.4051	0.001**

\*\*\*, \*\* and \* indicate statistical significance at the 1, 5 and 10% levels respectively

score matching (Rosenbaum and Rubin 1983) in which the predicted probability (propensity score) of being selected as a SSM-supervised bank is computed by estimating a simple probit model. We use bank capitalization (ETA) and profitability (ROA) to match SSMsupervised banks with nationally supervised banks using the Kernel matching approach (Heckman et al., 1998).<sup>18</sup> The results of the propensity score matching difference-in-differences estimation are reported in Table 10 and are consistent with those of our baseline models, supporting the robustness of our results. In Table 11, we compare the differences between treated and untreated banks for these two variables. It emerges that these differences are statistically insignificant, further supporting the effectiveness of our propensity score matching methodology. The second potential shortcoming of the DiD methodology

<sup>&</sup>lt;sup>18</sup> We select the variables ETA and ROA to match treated and untreated banks for two main reasons: First, capitalization and profitability are considered amongst the most important determinants of risk disclosure in the banking industry (Helbok and Wagner, 2006; Linsley et al., 2006). Second, the selection of additional control variables in the propensity score matching regression would lead to a significant decrease in the number of observations, thereby undermining the generalizability of our findings to the whole population. According to Shipman et al. (2017), this problem should be avoided when performing propensity score matching.

Variables	(1)	(2)	(3)	(4)
	Risk manage- ment disclosure	Risk exposure	References to the regulatory framework	Reassuring disclosure
BU	0.3212***	0.2236***	0.2551***	0.2094***
	(0.0079)	(0.0018)	(0.0021)	(0.0143)
Treat	0.1489***	0.2673***	0.0157	0.1843
	(0.0200)	(0.0132)	(0.0252)	(0.1238)
BU * Treat	-0.0878***	$-0.0170^{***}$	0.0742***	-0.0376***
	(0.0062)	(0.0058)	(0.0232)	(0.0053)
Observations	450	450	450	450
R-squared	0.313	0.430	0.389	0.317
Number of id	225	225	225	225
Robust standard errors	Yes	Yes	Yes	Yes
Clusters	Yes	Yes	Yes	Yes
Country fixed effects	Yes	Yes	Yes	Yes
Controls	Yes	Yes	Yes	Yes

 Table 12
 Addressing serial correlation concerns



Fig. 1 Parallel trend-Risk management category

is that it might suffer from serial correlation issues (Bertrand et al. 2004). We address this concern by collapsing the time series information into the pre- and post- 2014 period, taking the means of the data of these two time periods separately, and then repeating the DiD regression at the averaged level.<sup>19</sup> The results are reported in Table 12 and are qualitatively

<sup>&</sup>lt;sup>19</sup> This procedure for dealing with potential serial correlation is suggested by Bertrand et al. (2004).



Fig. 2 Parallel trend-Risk exposure category



**Category: Regulatory framework**  $\rho = 0.9$ 

Fig. 3 Parallel trend—Regulatory framework

unchanged from our baseline. Hence, our results are still robust after controlling for serial correlation issues.

Fifth, in order to guarantee that the results of our difference-in-differences strategy are reliable and not affected by other concurrent policies, the treatment and the control group must satisfy the parallel trend assumption in the pre-treatment period. Figures 1,2, 3, 4 show the correlation coefficients of the of the mean values of the disclosure indexes of SSM supervised banks and nationally supervised entities in the pre-treatment period. The correlation coefficients are high for all disclosure categories (greater or equal to 0.7) and visual inspection indicates that the trends of the disclosure indexes are parallel. Thus,



Fig. 4 Parallel trend—Reassurance

the control group of nationally supervised banks appears to be a valid counterfactual of the treatment group (SSM supervised banks).<sup>20</sup>In order to formally test the parallel trend assumption, we draw upon the approach proposed by Frankel et al. (2020) and Ahmed et al. (2020). Specifically, we added dummy variables that identify each year before the establishment of the BU, and the interactions between these indicator variables and the variable Treat. Formally, we used the following model:

$$DI_{Cark_{ijt}} = \beta_1 (BU * Treat_{ijt}) + \beta_2 \cdot dummy_2012 + \beta_3 \cdot dummy_2013 + \beta_4 \cdot dummy_2014 + \beta5 (dummy_2012 * Treat_{ijt}) + \beta_6 (dummy_2013 * Treat_{ijt}) + \beta_7 (dummy_2014 * Treat_{ijt}) + \gamma Con + \delta CFE + \varepsilon_{ijt}$$

$$(4)$$

where dummy\_2012, dummy\_2013 and dummy\_2014 are the dummy variables that identify years 2012, 2013 and 2014 respectively. The results of this test are reported in Table 13. First, for all parameters (with one single exception in column 1) there is no significant differential trend between treated and untreated banks before the establishment of the BU. Second, in order to compare the differences between treated and untreated banks before and after the banking union, we perform t-tests to compare the magnitude of the parameter associated to the interaction dummy BU\*Treat ( $\beta_1$ ), with the average of  $\beta_5$ ,  $\beta_6$  and  $\beta_7$ . The results are reported at the bottom of Table 13. They are all statistically different at conventional levels, suggesting that there is a significant differential decrease in disclosure for SSM supervised banks after the establishment of the BU. These results further support the robustness of our baseline model.

<sup>&</sup>lt;sup>20</sup> We carried out, but do not report, two further robustness tests that support the reliability of our results. First, we excluded non-eurozone countries from our analysis, and second, we computed the regression estimates using only the words 'risk' and 'uncertain', as they are considered particularly important when it comes to create a quantitative measure of language (Tetlock.et al. 2008). The results are available from the authors upon request.

Variables	(1)	(2)	(3)	(4)
	Risk management	Risk exposure	Regulatory framework	Reassurance
BU*Treat ( $\beta_1$ )	-0.2028***	-0.1896***	0.1542***	-0.2419***
	(0.0340)	(0.0401)	(0.0399)	(0.0251)
Dummy_2012 ( $\beta_2$ )	0.0271**	0.0276*	-0.0060	-0.0113
	(0.0121)	(0.0147)	(0.0137)	(0.0167)
Dummy_2013 (β <sub>3</sub> )	-0.0202**	-0.0731	0.0171	-0.0952***
	(0.0087)	(0.0728)	(0.0138)	(0.0104)
Dummy_2014 (β <sub>4</sub> )	-0.0106	-0.0548***	0.0737	-0.0045
	(0.0160)	(0.0176)	(0.0725)	(0.0122)
Dummy_2012*Treat ( $\beta_5$ )	-0.0265	0.0245	-0.0918	0.0646
	(0.0177)	(0.1051)	(0.0986)	(0.0324)
Dummy_2013*Treat ( $\beta_6$ )	-0.0103	-0.0245	-0.0157	-0.0350
	(0.0140)	(0.0251)	(0.0658)	(0.0360)
Dummy_2014*Treat ( $\beta_7$ )	0.1290*	-0.0387	0.1046	-0.0311
· · · · · ·	(0.0750)	(0.0340)	(0.0727)	(0.0343)
$\beta_1 - (\beta_5 + \beta_6 + \beta_7) / 3$	-0.2335**	-0.1767***	0.1552*	-0.2414**
$H_0: \beta_1 - (\beta_5 + \beta_6 + \beta_7) / 3$	t = -5.7951	t = -11.3044	t = 3.3240	t = -9.0776
Observations	1,209	1,209	1,209	1,209
R-squared	0.347	0.415	0.377	0.298
Number of id	225	225	225	225
Robust standard errors	Yes	Yes	Yes	Yes
Clusters	Yes	Yes	Yes	Yes
Country fixed effects	Yes	Yes	Yes	Yes
Controls	Yes	Yes	Yes	Yes

 Table 13
 Test of parallel trend assumption

The dependent variable is a bank risk disclosure index for different categories of bank risk. Robust standard errors in parenthesis. \*\*\*, \*\* and \* indicate statistical significance at the 1, 5 and 10% levels respectively

Finally, we examine the robustness of our results to the inclusion in Eq. (3) of corporate governance variables, which might affect the results for two reasons. First, by adding additional control variables, we reduce possible concerns related to omitted variable bias that could affect our econometric model. Second, previous studies provide evidence that corporate governance plays an important role in bank risk taking (Beltratti and Stulz 2012; Minton et al 2014) and disclosure (Forker 1992). Hence, we extend our empirical model by adding the following variables: (i) the CEO's age in years (CEOage); (ii) CEO's gender (CEOgen), which is a dummy variable equal to 1 if the CEO is female and zero otherwise; (iii) whether the CEO also holds the post of chairman (CEOdual), which is a dummy variable equal to 1 if the CEO is also the chairman of the bank and zero otherwise; (iv) whether the bank has a two-tier executive board (2TS), which is a dummy variable equal to 1 if the board structure includes supervisors and managers and zero otherwise; and the natural logarithm of the number of senior executives (directors and managers) in the bank (SeniorExec). The sources of these variables are Bordex and Thomson Reuters Eikon. The results of the regressions are reported in Table 14. The signs and statistical significance of the coefficients on the main variable of interest (BU and BU\*Treat) are qualitatively unchanged from our baseline model, supporting the robustness of our results. As for

Variables	(1)	(2)	(3)	(4)
	Risk management	Risk exposure	Regulatory framework	Reassurance
BU	0.2341***	0.2318***	0.1527**	0.1577***
	(0.0404)	(0.0524)	(0.0723)	(0.0392)
Treat	0.1136**	0.2483*	0.0177	0.1346
	(0.0548)	(0.1270)	(0.1337)	(0.0825)
BU*Treat	-0.1774***	-0.1587***	0.1209**	-0.1831***
	(0.0374)	(0.0446)	(0.0613)	(0.0335)
ETA	-0.7743***	-0.8134***	0.1329	-0.0452
	(0.2260)	(0.3120)	(0.1460)	(0.3611)
ROA	-0.1297**	-0.1208***	-0.0261	-0.1176***
	(0.0617)	(0.0326)	(0.0550)	(0.0203)
LOANS	0.2444**	0.1333	0.3939***	0.5668**
	(0.1107)	(0.1470)	(0.1410)	(0.2638)
OBS	-0.0260	-0.0386	-0.2187***	-0.1541
	(0.0490)	(0.0634)	(0.0500)	(0.1019)
CUST	-0.3798***	-0.2859***	-0.1059	0.1128
	(0.1472)	(0.0716)	(0.1937)	(0.1195)
BIG4D	0.0261	0.1859	0.3642*	0.0095
	(0.0819)	(0.1426)	(0.1926)	(0.2120)
WBBD	0.1821**	0.1071*	0.1195***	0.0811
	(0.0717)	(0.0625)	(0.0255)	(0.0500)
NSAD	-0.3827	0.0353	-0.5051*	-0.0023
	(0.4087)	(0.1758)	(0.3051)	(0.2500)
CEOage	0.0123**	0.0034	0.0159**	0.0087*
	(0.0055)	(0.0050)	(0.0068)	(0.0049)
CEOgen	0.0915	0.1281	0.3403***	0.0573
	(0.1085)	(0.1451)	(0.1282)	(0.1780)
CEOdual	-0.1879**	-0.2720*	0.1099	0.1598
	(0.0772)	(0.1510)	(0.1747)	(0.2313)
2TS	0.3134	0.2905	0.1834	0.1178
	(0.3110)	(0.2527)	(0.2368)	(0.2155)
SeniorExec	0.1951***	0.2182***	0.1022	0.1169**
	(0.0757)	(0.0727)	(0.1077)	(0.0518)
Observations	1,209	1,209	1,209	1,209
R-squared	0.347	0.415	0.373	0.298
Number of id	225	225	225	225
Robust standard errors	Yes	Yes	Yes	Yes
Clusters	Yes	Yes	Yes	Yes
Country fixed effects	Yes	Yes	Yes	Yes

 Table 14
 Effects of banking Union on Bank disclosure—including corporate governance variables

the effects of the corporate governance variables, CEOage has a positive and statistically significant effect in three out of four regressions, in line with previous studies that found

Variables	(1)	(2)	(3)	(4)
Variables	Risk management	Risk exposure	Regulatory framework	Reassurance
BU	0.2493***	0.2460***	0.1594**	0.1669***
	(0.0409)	(0.0527)	(0.0724)	(0.0389)
Treat	0.1225**	0.2578**	0.0219	0.1403*
	(0.0550)	(0.1256)	(0.1334)	(0.0819)
BU*Treat	-0.1885***	-0.1700***	0.1157*	-0.1892***
	(0.0337)	(0.0436)	(0.0604)	(0.0326)
BU*ROA	-2.1118***	-1.9856***	-0.9726***	-1.3012**
	(0.5835)	(0.5402)	(0.2556)	(0.5454)
Observations	1,209	1,209	1,209	1,209
R-squared	0.345	0.415	0.372	0.297
Number of id	225	225	225	225
Robust standard errors	Yes	Yes	Yes	Yes
Clusters	Yes	Yes	Yes	Yes
Country fixed effects	Yes	Yes	Yes	Yes
Controls	Yes	Yes	Yes	Yes

Table 15 Interaction between BU and ROA

that the age of the CEO is an important determinant of disclosure (Marquez-Illescas et al. 2019). These results show that older CEOs provide higher levels of disclosure, because of their higher experience that enables them to adopt more effective and risk-oriented disclosure practices. CEOgen is positively related with disclosure, showing that women CEOs are more prone to provide higher levels of disclosure. This supports recent research suggesting that female CEOs are more risk averse (Faccio et al. 2016; Martin et al. 2009). However, it is necessary to be cautious in the interpretation of this effect, because it is statistically significant only in one out of four regressions. CEOdual is negatively and significantly related with two risk disclosure variables. This result shows that when CEOs act also as chairmen in banking institutions, the levels of disclosure are lower. This finding is in line with the agency theory (Jensen and Meckling 1976), given that CEO duality implies higher levels of power for the CEO, who may decide to provide low levels of disclosure, because he/she is not interested in reducing information asymmetries with their shareholders. Lastly, SeniorExec is positively and significantly associated with disclosure in three out of four regressions. suggesting that more experienced directors and managers contribute to increasing the level of disclosure (Adawi and Rwegasira 2011).

We also carry out additional analyses taking into consideration interaction terms between our variables of interest and selected explanatory variables for two main reasons. First, to test whether these variables may affect the relationship between BU and disclosure. Second, because, by including additional variables in our regression model, we can further test the robustness of our baseline results.

Specifically, we investigate into the following aspects: (i) how bank profitability influences the relationship between BU and disclosure; (ii) how the location of banks in GIPSI (Greece, Italy, Portugal, Spain and Ireland) countries influences the relationship between BU and disclosure; (iii) how CEO duality (i.e. the CEO is also the Chairman of the

Variables	(1)	(2)	(3)	(4)
	Risk management	Risk exposure	Regulatory framework	Reassurance
BU	0.2134***	0.2152***	0.1503**	0.1488***
	(0.0343)	(0.0447)	(0.0744)	(0.0343)
Treat	0.1199*	0.2546**	0.0183	0.1372*
	(0.0640)	(0.1207)	(0.1343)	(0.0781)
BU*Treat	-0.2000***	-0.1790***	0.1186*	-0.1911***
	(0.0370)	(0.0395)	(0.0640)	(0.0354)
BU*GIPSI	0.1292*	0.1017**	0.0143	0.0553
	(0.0669)	(0.0498)	(0.0288)	(0.0462)
Observations	1,209	1,209	1,209	1,209
R-squared	0.348	0.414	0.373	0.298
Number of id	225	225	225	225
Robust standard errors	Yes	Yes	Yes	Yes
Clusters	Yes	Yes	Yes	Yes
Country fixed Effects	Yes	Yes	Yes	Yes
Controls	Yes	Yes	Yes	Yes

#### Table 16 Interaction between BU and GIPSI

Robust standard errors in parenthesis. \*\*\*, \*\* and \* indicate statistical significance at the 1, 5 and 10% levels respectively

(-)	(4)
sure Regulatory framework	Reassurance
* 0.1524**	0.1578***
(0.0723)	(0.0393)
0.0155	0.1320
(0.1327)	(0.0850)
** 0.1322**	-0.1549***
(0.0550)	(0.0422)
* -0.1414	-0.3720**
(0.1524)	(0.1877)
1,209	1,209
0.371	0.295
225	225
Yes	Yes
	(0)           sure         Regulatory framework           0.1524**         (0.0723)           0.0155         (0.1327)           **         0.1322**           (0.0550)         *           -0.1414         (0.1524)           1,209         0.371           225         Yes           Yes         Yes           Yes         Yes           Yes         Yes

#### Table 17 Interaction between CEO duality and BU\*Treat

Robust standard errors in parenthesis. \*\*\*, \*\* and \* indicate statistical significance at the 1, 5 and 10% levels respectively

financial institution) affects the relationship between the interaction dummy ( $BU^*Treat$ ) and bank disclosure. The results are reported in the Table 15, 16 and 17. Table 15 shows that the interaction term between BU and ROA is negatively and significantly associated with each of the four dependent variables. This finding means that the relationship between BU and disclosure is influenced by bank profitability. In particular, the positive effect of the BU on bank disclosure is stronger for less profitable banks and weaker for highly profitable banks. This result is important because the disclosures provided by less profitable banks are particularly relevant for investors and stakeholders that are interested in understanding the reasons of their low level of profitability. Table 16 reports the effect of the interaction between the BU indicator variable and a dummy variable that identifies the banks that are located in the GIPSI countries. Shambaugh (2012) uses this acronym to indicate the five most troubled economies of the Eurozone, namely Greece, Italy, Portugal, Spain, Ireland. Previous studies have argued that the effect of the BU in GIPSI countries might be different from the other European countries because of their high levels of sovereign debt (Avignone et al. 2020). In order to investigate into this aspect, we analyze if the location of banks in GIPSI countries affect the way the BU has influenced bank disclosure. The results show that the interaction between BU and the GIPSI dummy variable is positively and significantly associated with two risk disclosure categories. This finding support the idea that the effect of the banking union has been different for these countries, and in particular the positive influence of the BU on disclosure has been stronger in these countries. Similarly to the results obtained in the previous regression, also in this case the disclosure has increased for those banks that were in a more difficult situation, and in particular for those banks that are located in troubled countries characterized by high levels of sovereign debt.<sup>21</sup> Finally, in Table 17 we report on the interaction between BU, Treat and CEO duality. The results show that for those banks in which the CEO is also the Chairman of the bank (CEO duality), the negative effect of the banking union on the disclosure of SSM supervised banks is stronger. This effect is statistically significant for two out of four disclosure categories. This finding is consistent with previous literature that shows that CEO duality has a negative effect on disclosure (Gul and Leung 2004). The hypothesis that concentrated decision making power, and consequently CEO duality, may negatively influence board's oversight, including disclosure policies is grounded in agency theory (Fama and Jensen 1983). This is because CEO duality and its associated concentrated decision making power could reduce the ability of the board to exercise effective control. Thus, this mechanism played an important role in determining the effect of BU on the disclosure practices of SSM supervised banks. In Tables 15, 16 and 17, the results of the main variables of interest (BU and BU\*TREAT) are qualitatively unchanged from our baseline regressions and they are still statistically significant at conventional levels. These findings further validate the robustness of our baseline results.

<sup>&</sup>lt;sup>21</sup> We build upon the evidence that the effect of the BU is different in the most indebted European countries by performing an additional analysis on the least indebted countries. The results (not reported) show that the negative effect of the banking union on the risk disclosure of SSM supervised banks is less strong in the least indebted European countries. This finding provides additional evidence that the level of sovereign debt of the countries has played an important role in shaping the relationship between the BU and bank disclosure practices. The results of these analyses are available from the authors upon request.

# 6 Conclusions

This paper investigated the impact of BU on bank risk disclosure in a sample of 225 European banks, 75 of which were supervised under the SSM once BU was introduced. It employed content analysis techniques to construct a dictionary designed to assess four categories of disclosure in bank financial statements together with a Difference-in-Differences estimation methodology in which banks supervised under the SSM act as the treatment group and banks supervised by the national supervisor act as the control group. The main hypotheses underlying the analysis were that the onset of BU would increase risk disclosure by all banks and that the additional requirements of the SSM would increase risk disclosure by SSM supervised banks (treated entities) relative to that by other banks. The paper reports two key results that are robust to a battery of tests. First, BU had a positive impact on bank disclosure overall—i.e., bank disclosure in a number of categories increased following the onset of BU. Second, in line with our second research hypothesis, there is evidence that disclosure by SSM-supervised banks worsened relative to that of nationally supervised banks. The latter result likely reflects the shortcoming of a two-tier supervisory system in which the SSM supervisor-the ECB-is dependent on national regulators for the supply of information. This arrangement likely gives rise to inefficiencies with respect to the speed and completeness of the information flow between SSM supervised banks and the ECB, especially as the interest of the ECB and the national regulators are unlikely to always converge. These inefficiencies, already identified by other authors (Carletti et al. 2015, 2021; Gros and Schoenmaker 2014), might affect bank risk disclosure practices, resulting in the lower level of disclosure observed in our empirical analysis. In additional analyses, we also find that the overall positive effect of the BU on bank disclosure is stronger for less profitable banks and in the most troubled economies of the Eurozone, while the negative effect on SSM supervised banks is stronger if bank CEOs act also as Chairman. These results show that bank specific and country characteristics play an important role in determining the impact of banking supervision on bank disclosure practices.

We acknowledge that our paper is potentially affected by a caveat: the results might be partially driven by other competing regulations occurred in 2014, different from the establishment of the BU and its associated rules. However, to our best knowledge, the only regulation that could affect our results is the already mentioned 2014 Basel III Pillar 3 disclosures requirement. In this regard, it is important to notice that this regulation has imposed disclosure requirements exclusively in the Pillar 3 disclosure report, which is provided separately from the annual financial reports of European banks (Scannella and Polizzi 2018). Given that we did not include the Pillar 3 disclosure report in our content analysis, the results of the empirical investigation should not be affected by this regulation. The main policy implication of our results is that efforts are necessary to improve disclosure by SSM-supervised banks to the same extent as that of nationally supervised banks. One way to achieve this goal would be have SSM-supervised banks report directly to the ECB.

Acknowledgements We thank Giuseppe Avignone, Danilo Conte, Antonio D'Antoni, Alessio Reghezza, Laura Santucci and two anonymous referees for their valuable suggestions and advices. We also thank the participants of the following conferences: 28th Annual Conference on Pacific Basin Finance, Economics, Accounting and Management & 14th NCTU International Finance Conference, ADEIMF Summer Conference 2020, Wolpertinger Conference 2019.

**Funding** Open access funding provided by Università degli Studi di Palermo within the CRUI-CARE Agreement. No funding was received for conducting this study.

#### Declarations

**Conflict of interest** The authors have no conflict of interest to declare that are relevant to the content of this article.

**Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

# References

- Abraham S, Cox P (2007) Analysing the determinants of narrative risk information in UK FTSE 100 annual reports. The Br Account Rev 39:227–248. https://doi.org/10.1016/j.bar.2007.06.002
- Adawi M, Rwegasira K (2011) Corporate boards and voluntary implementation of best disclosure practices in emerging markets: evidence from the UAE listed companies in the Middle East. Int J of Discl and Gov 8:272–293. https://doi.org/10.1057/jdg.2011.5
- Agarwal S, Lucca D, Seru A, Trebbi F (2014) Inconsistent regulators: evidence from banking. The Q J of Econ 129:889–938. https://doi.org/10.1093/qje/qju003
- Al Lawati H, Hussainey K, Sagitova R (2021) Disclosure quality vis-à-vis disclosure quantity: does audit committee matter in Omani financial institutions? Rev of Quant Finan and Account, Forthcoming. https://doi.org/10.1007/s11156-020-00955-0
- Al-Hadi A, Hasan MM, Habib A (2016) Risk committee, firm life cycle, and market risk disclosures. Corp Gov: an Int Rev 24:145–170. https://doi.org/10.1111/corg.12115
- Ahmed AS, Li Y, Xu N (2020) Tick size and financial reporting quality in small-cap firms: evidence from a natural experiment. J of Account Res 58:869–914. https://doi.org/10.1111/1475-679X.12331
- Andrés P, Polizzi S, Scannella E, Suárez N (2021) Corruption-Related Disclosure in the Banking Industry: Evidence From GIPSI Countries. European Corporate Governance Institute (ECGI) finance working paper series 745/2021.
- Avgouleas E (2009) The global financial crisis, behavioural finance and financial regulation: in search of a new orthodoxy. J of Corporat Law Stud 9:23–59. https://doi.org/10.1080/14735970.2009.11421534
- Avignone G, Altunbas Y, Polizzi S, Reghezza A (2020) Centralised or decentralised banking supervision? evidence from European banks. J of Int Money and Financ 110:1–14. https://doi.org/10.1016/j.jimon fin.2020.102264
- Balakrishnan K, Billings MB, Kelly B, Ljungqvist A (2014) Shaping liquidity: on the causal effects of voluntary disclosure. The J of Financ 69:2237–2278. https://doi.org/10.1111/jofi.12180
- Ball R, Shivakumar L (2005) Earnings quality in UK private firms: comparative loss recognition timeliness. J of Account and Econ 39:83–128. https://doi.org/10.1016/j.jacceco.2004.04.001
- Ball R, Robin A, Wu JS (2003) Incentives versus standards: properties of accounting income in four East Asian countries. J of Account and Econ 36:35–270. https://doi.org/10.1016/j.jacceco.2003.10.003
- Barakat A, Hussainey K (2013) Bank governance, regulation, supervision, and risk reporting: evidence from operational risk disclosures in European banks. Int Rev of Financ Anal 30:254–273. https://doi. org/10.1016/j.irfa.2013.07.002
- Barth ME, Israeli D (2013) Disentangling mandatory IFRS reporting and changes in enforcement. J of Account and Econ 56:178–188. https://doi.org/10.1016/j.jacceco.2013.11.002
- Beattie V (2014) Accounting narratives and the narrative turn in accounting research: issues, theory, methodology, methods and a research framework. The Br Account Rev 46:111–134. https://doi.org/10. 1016/j.bar.2014.05.001
- Beltratti A, Stulz RM (2012) The credit crisis around the globe: why did some banks perform better? J of Finan Econ 105:1–17. https://doi.org/10.1016/j.jfineco.2011.12.005

- Benczur P, Cannas G, Cariboni J, Di Girolamo F, Maccaferri S, Giudici MP (2017) Evaluating the effectiveness of the new EU bank regulatory framework: a farewell to bail-out? J of Financ Stab 33:207–223. https://doi.org/10.1016/j.jfs.2016.03.001
- Berger AN, Kick T, Schaeck K (2014) Executive board composition and bank risk taking. J of Corp Financ 28:48–65. https://doi.org/10.1016/j.jcorpfin.2013.11.006
- Berger AN, Turk-Ariss R (2015) Do depositors discipline banks and did government actions during the recent crisis reduce this discipline? an international perspective. J of Financ Serv Res 48:103–126. https://doi.org/10.1007/s10693-014-0205-7
- Bertrand M, Duflo E, Mullainathan S (2004) How much should we trust differences-in-differences estimates? The Q J of Econ 119:249–275. https://doi.org/10.1162/003355304772839588
- Bolton P, Ochmke M (2019) Bank resolution and the structure of global banks. The Rev of Financ Stud 32:2384–2421. https://doi.org/10.1093/rfs/hhy123
- Botosan CA, Plumlee MA (2002) A re-examination of disclosure level and the expected cost of equity capital. J of Account Res 40:21–40. https://doi.org/10.1111/1475-679X.00037
- Botosan CA (1997) Disclosure level and the cost of equity capital. Account Rev 72:323-349
- Bowen RM, DuCharme L, Shores D (1995) Stakeholders' implicit claims and accounting method choice. J of Account and Econ 20:255–295. https://doi.org/10.1016/0165-4101(95)00404-1
- Boyd B (1990) Corporate linkages and organizational environment: a test of the resource dependence model. Strateg Manag J 11:419–430. https://doi.org/10.1002/smj.4250110602
- Brown SV, Tucker JW (2011) Large-sample evidence on firms' year-over-year MD&A modifications. J of Account Res 49:309–346. https://doi.org/10.1111/j.1475-679X.2010.00396.x
- Burgstahler D, Dichev I (1997) Earnings management to avoid earnings decreases and losses. J of Account and Econ 24:99–126. https://doi.org/10.1016/S0165-4101(97)00017-7
- Burks JJ, Cuny C, Gerakos J, Granja J (2018) Competition and voluntary disclosure: evidence from deregulation in the banking industry. Rev of Account Stud 23:1471–1511. https://doi.org/10.1007/ s11142-018-9463-1
- Bushman RM, Hendricks BE, Williams CD (2016) Bank competition: measurement, decision-making, and risk-taking. J of Account Res 54:777–826. https://doi.org/10.1111/1475-679X.12117
- Carboni M, Fiordelisi F, Ricci O, Lopes FSS (2017) Surprised or not surprised? the investors' reaction to the comprehensive assessment preceding the launch of the banking union. J of Bank and Financ 74:122–132. https://doi.org/10.1016/j.jbankfin.2016.11.004
- Carletti E, Dell'Ariccia G, Marquez R (2015) Supervisory incentives in a banking union. IMF Working Paper 16/186.
- Carletti E, Dell'Ariccia G, Marquez R (2021) Supervisory incentives in a banking union. Manag Sci 67:455–470. https://doi.org/10.1287/mnsc.2019.3448
- Carlson M, Wheelock DC (2018) Did the founding of the Federal Reserve affect the vulnerability of the interbank system to contagion risk? J of Money Credit and Bank 50:1711–1750. https://doi.org/10. 11111/jmcb.12520
- Chen Q, Vashishtha R (2017) The effects of bank mergers on corporate information disclosure. J of Account and Econ 64:56–77. https://doi.org/10.1016/j.jacceco.2017.05.003
- Cheng CA, Collins D, Huang HH (2006) Shareholder rights, financial disclosure and the cost of equity capital. Rev of Quant Financ and Account 27:175–204. https://doi.org/10.1007/ s11156-006-8795-2
- Cho CH, Laine M, Roberts RW (2015) Organized hypocrisy, organizational façades, and sustainability reporting. Account, Organ and Soc 40:78–94. https://doi.org/10.1016/j.aos.2014.12.003
- Cohen J, Krishnamoorthy G, Wright A (2017) Enterprise risk management and the financial reporting process: the experiences of audit committee members, CFOs, and external auditors. Contemp Account Res 34:1178–1209. https://doi.org/10.1111/1911-3846.12294
- Cordella T, Yeyati EL (1998) Public disclosure and bank failures. IMF Econ Rev 45:110–131. https:// doi.org/10.2307/3867331
- Costello AM, Granja J, Weber J (2018) Do strict regulators increase the transparency of banks? J of Account Res. https://doi.org/10.1111/1475-679X.12255
- Cumming JA, Wooff DA (2007) Dimension reduction via principal variables. Comput Stat Data Anal 52:550–565. https://doi.org/10.1016/j.csda.2007.02.012
- Draghi M (2018) The benefits of European supervision. Speech at the Autorité de Contrôle Prudentiel et de Résolution Conference on Financial Supervision. https://www.ecb.europa.eu/press/key/date/ 2018/html/ecb.sp180918.en.html, Accessed 27 February 2020.
- Eaton TV, Nofsinger JR, Weaver DG (2007) Disclosure and the cost of equity in international crosslisting. Rev of Quant Financ and Account 29:1–24. https://doi.org/10.1007/s11156-007-0024-0

- Easley D, O'Hara M (2004) Information and the cost of capital. J Financ 59:1553–1583. https://doi.org/ 10.1111/j.1540-6261.2004.00672.x
- Eng LL, Mak YT (2003) Corporate governance and voluntary disclosure. J Account Public Policy 22:325–345. https://doi.org/10.1016/S0278-4254(03)00037-1
- Faccio M, Marchica M-T, Mura R (2016) CEO gender, corporate risk-taking, and the efficiency of capital allocation. J Corp Financ 39:192–209
- Fama E, Jensen MC (1983) Separation of ownership and control. J Econ Law 26:301–325. https://doi.org/ 10.1086/467037
- Fiordelisi F, Ricci O, Stentella Lopes FS (2017) The unintended consequences of the launch of the single supervisory mechanism in Europe. J Financ Quant Anal 52:2809–2836. https://doi.org/10.1017/ S0022109017000886
- Fitch TP (2018) Dictionary of Banking Terms, 7th. Barron's Educational Series, New York
- Flannery MJ (2001) The faces of "market discipline." J Financ Serv Res 20:107–119. https://doi.org/10. 1023/A:1012455806431
- Flannery MJ, Kwan SH, Nimalendran M (2013) The 2007–2009 financial crisis and bank opaqueness. J Financ Intermed 22:55–84. https://doi.org/10.1016/j.jfi.2012.08.001
- Forker JJ (1992) Corporate governance and disclosure quality. Account Bus Res 22:111-124
- Frankel R, Kim BH, Ma T, Martin X (2020) Bank monitoring and financial reporting quality: the case of accounts receivable-based loans. Contemp Account Res 37:2120–2144. https://doi.org/10.1111/1911-3846.12595
- Goncharenko R, Hledik J, Pinto R (2018) The dark side of stress tests: negative effects of information disclosure. J of Financ Stab 37:49–59. https://doi.org/10.1016/j.jfs.2018.05.003
- Gorton G, Ordonez G (2014) Collateral crises. American Econ Rev 104:343–78. https://https://doi.org/10. 1257/aer.99.2.567
- Gorton G (2009) Information, liquidity, and the (ongoing) panic of 2007. American Econ Rev 99:567–572. https://doi.org/10.1257/aer.99.2.567
- Gros D, Schoenmaker D (2014) European deposit insurance and resolution in the Banking Union. J Common Mark Stud 52:529–546. https://doi.org/10.1111/jcms.12124
- Gul FA, Leung S (2004) Board leadership, outside directors' expertise and voluntary corporate disclosures. J Account Public Policy 23:351–379. https://doi.org/10.1016/j.jaccpubpol.2004.07.001
- Gunther JW, Moore RR (2003) Loss underreporting and the auditing role of bank exams. J Financ Intermed 12:153–177. https://doi.org/10.1016/S1042-9573(03)00015-9
- Hartigan JA, Wong MA (1979) Algorithm AS 136: A k-means clustering algorithm. J Royal Stat Soc Series C (appl Stat) 28:100–108. https://doi.org/10.2307/2346830
- Heckman J, Ichimura H, Smith J, Todd P (1998) Characterizing selection bias using experimental data. Econometrica 66:1017–1098. https://doi.org/10.2307/2999630
- Helbok G, Wagner C (2006) Determinants of operational risk reporting in the banking industry. The J of Risk 9:49–74. https://doi.org/10.2139/ssrn.425720
- Hertzberg A, Liberti JM, Paravisini D (2011) Public information and coordination: evidence from a credit registry expansion. J Financ 66:379–412. https://doi.org/10.1111/j.1540-6261.2010.01637.x
- Holthausen C, Rønde T (2004) Cooperation in international bank supervision. ECB working paper No. 316.
- Hüser AC, Halaj G, Kok C, Perales C, van der Kraaij AF (2017) The systemic implications of bail-in: a multi-layered network approach. J Financ Stab 38:81–97. https://doi.org/10.1016/j.jfs.2017.12.001
- Illia L, Sonpar K, Bauer MW (2014) Applying co-occurrence text analysis with ALCESTE to studies of impression management. Br J Man 25:352–372. https://doi.org/10.1111/j.1467-8551.2012.00842.x
- Jensen MC, Meckling WH (1976) Theory of the firm: managerial behavior, agency costs and ownership structure. J Financ Econ 3:305–360. https://doi.org/10.1016/0304-405X(76)90026-X
- Kearney C, Liu S (2014) Textual sentiment in finance: a survey of methods and models. Int Rev Financ Anal 33:171–185. https://doi.org/10.1016/j.irfa.2014.02.006
- Kleymenova A, Zhang L (2019) The impact of banking regulation on voluntary disclosures: Evidence from the Dodd-Frank act. Chicago Booth Research Paper 19–08.
- Kothari SP, Li X, Short JE (2009) The effect of disclosures by management, analysts, and business press on cost of capital, return volatility, and analyst forecasts: a study using content analysis. Account Rev 84:1639–1670. https://doi.org/10.2308/accr.2009.84.5.1639
- Krink T, Paterlini S, Resti A (2007) Using differential evolution to improve the accuracy of bank rating systems. Comput Stat Data Anal 52:68–87. https://doi.org/10.1016/j.csda.2007.02.011
- Krippendorff K (2004) Content analysis: an introduction to its methodology, 2nd edn. Sage, Thousand Oaks
- Landsman WR (2006) Fair Value Accounting for Financial Instruments: Some Implications for Bank Regulation. BIS Working Paper 209.

- Larcker DF, Richardson SA, Tuna IR (2007) Corporate governance, accounting outcomes, and organizational performance. Account Rev 82:963–1008. https://doi.org/10.2308/accr.2007.82.4.963
- Law J, Smullen J (2018) A dictionary of finance and banking, 4th edn. Oxford University Press, Oxford
- Li S (2010) Does mandatory adoption of International Financial Reporting Standards in the European Union reduce the cost of equity capital? Account Rev 85:607–636
- Linsley PM, Shrives PJ (2006) Risk reporting: a study of risk disclosures in the annual reports of UK companies. Br Account Rev 38:387–404. https://doi.org/10.1016/j.bar.2006.05.002
- Linsley PM, Shrives PJ, Crumpton M (2006) Risk disclosure: an exploratory study of UK and Canadian banks. J of Bank Reg 7:268–282. https://doi.org/10.1057/palgrave.jbr.2350032
- Loughran T, McDonald B (2011) When is a liability not a liability? Textual analysis, dictionaries, and 10-Ks. The J of Financ 66:35–65. https://doi.org/10.1111/j.1540-6261.2010.01625.x
- Ludvigson SC, Ng S (2007) The empirical risk–return relation: a factor analysis approach. J Financ Econ 83:171–222. https://doi.org/10.1016/j.jfineco.2005.12.002
- Marquez-Illescas G, Zebedee AA, Zhou L (2019) Hear me write: does CEO narcissism affect disclosure? J of Bus Ethics 159:401–417. https://doi.org/10.1007/s10551-018-3796-3
- Martin AD, Nishikawa T, Williams MA (2009) CEO gender: effects on valuation and risk. Quart J Finan Account 48:23–40
- Mester LJ (2017) The nexus of macroprudential supervision, monetary policy, and financial stability. J Financ Stab 30:177–180. https://doi.org/10.1016/j.jfs.2017.07.003
- Miihkinen A (2012) What drives quality of firm risk disclosure?: the impact of a national disclosure standard and reporting incentives under IFRS. Int J Account 47:437–468. https://doi.org/10.1016/j.intacc. 2012.10.005
- Morgan DP (2002) Rating banks: risk and uncertainty in an opaque industry. American Econ Rev 92:874– 888. https://doi.org/10.1257/00028280260344506
- Minton BA, Taillard JP, Williamson R (2014) Financial expertise of the board, risk taking, and performance: Evidence from bank holding companies. J of Financ and Quant Anal 49:351–380. https://www.jstor.org/stable/43303949
- Nier E, Baumann U (2006) Market discipline, disclosure and moral hazard in banking. J Financ Intermed 15:332–361. https://doi.org/10.1016/j.jfi.2006.03.001
- Nielsen B, Smeets S (2018) The role of the EU institutions in establishing the banking union. collaborative leadership in the EMU reform process. J Eur Public Policy 25:1233–1256. https://doi.org/10.1080/ 13501763.2017.1285342
- Nissim D (2003) Reliability of banks' fair value disclosure for loans. Rev Quant Financ Account 20:355– 384. https://doi.org/10.1023/A:1024072317201
- O'Donovan G (2002) Environmental disclosures in the annual report: Extending the applicability and predictive power of legitimacy theory. Account Audit Account J 15:344–371. https://doi.org/10.1108/ 09513570210435870
- Östberg P (2006) Disclosure, investment and regulation. J Financ Intermed 15:285–306
- Pancotto L, ap Gwilym O, Williams J (2019) The European Bank Recovery and Resolution Directive: A Market Assessment. J of Financ Stab 44:1–16
- Petersen MA (2009) Estimating standard errors in finance panel data sets: comparing approaches. Rev Financ Stud 22:435–480. https://doi.org/10.1093/rfs/hhn053
- Richardson AJ, Welker M (2001) Social disclosure, financial disclosure and the cost of equity capital. Account Organ Soc 26:597–616. https://doi.org/10.1016/S0361-3682(01)00025-3
- Rosenbaum PR, Rubin DB (1983) The central role of the propensity score in observational studies for causal effects. Biometrika 70:41–55. https://doi.org/10.1093/biomet/70.1.41
- Rutherford D (2013) Routledge dictionary of economics, 3rd edn. Routledge, London
- Sahin C, De Haan J (2016) Market reactions to the ECB's comprehensive assessment. Econ Lett 140:1–5. https://doi.org/10.1016/j.econlet.2015.12.011
- Sáiz MC, Azofra SS, Olmo BT (2019) The single supervision mechanism and contagion between bank and sovereign risk. J Regul Econ 55:67–106. https://doi.org/10.1007/s11149-018-09373-6
- Shambaugh CJ (2012) The Euro's three crises. Brookings Pap Econ Activ 3:1–54
- Scannella E, Polizzi S (2018) Market risk disclosure in banking: an empirical analysis on four global systemically important European banks. J Bank Reg 19:87–100. https://doi.org/10.1057/s41261-017-0039-y
- Scannella E, Polizzi S (2021) How to measure bank credit risk disclosure? testing a new methodological approach based on the content analysis framework. J Bank Reg 22:73–95. https://doi.org/10.1057/ s41261-020-00129-x
- Schepens G (2016) Taxes and bank capital structure. J Financ Econ 120:585–600. https://doi.org/10.1016/j. jfineco.2016.01.015

Shim JK, Constas M (2001) Encyclopedic dictionary of international finance and banking. CRC Press, New York

- Shipman JE, Swanquist QT, Whited RL (2017) Propensity score matching in accounting research. Account Rev 92:213–244. https://doi.org/10.2308/accr-51449
- Sowerbutts R, Zimmerman P (2016) Market Discipline, Public Disclosure and Financial Stability. In: Haven E, Molyneux P, Wilson JOS, Fedotov S, Duygun M (eds) The Handbook of Post Crisis Financial Modeling. Palgrave Macmillan, London, pp 42–64
- Sowerbutts R, Zimmerman P, Zer I (2013) Banks' disclosure and financial stability. Bank England Quarterly Bulletin 53:326–445
- Tetlock PC, Saar-Tsechansky M, Macskassy S (2008) More than Words: Quantifying Language to Measure Firms' Fundamentals. J Financ 63:1437–1467. https://doi.org/10.1111/j.1540-6261.2008.01362.x
- Tröger TH (2014) The single supervisory mechanism: panacea or quack banking regulation? Eur Bus Organ Law Rev 14:449–497. https://doi.org/10.1017/S1566752914001244
- Weber RP (1990) Basic content analysis, 2nd edn. Sage, Beverly Hills
- White H (1980) A heteroskedasticity-consistent covariance matrix estimator and a direct test for heteroskedasticity. Econometrica 48:817–838.
- Woods M, Dowd K, Humphrey C (2008) Market risk reporting by the world's top banks: evidence on the diversity of reporting practice and the implications for international accounting harmonisation. Span Account Rev 11:9–42
- Zhang L, Ding S (2006) The effect of increased disclosure on cost of capital: evidence from China. Rev Quant Financ Account 27:383–401. https://doi.org/10.1007/s11156-006-0044-1

**Publisher's Note** Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.